## Format for ANSWERING REVIEWERS

October 08, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 5127-Review.docx).

Title: Helicobacter pylori Infection in Gastric Mucosa-Associated Lymphoid Tissue Lymphoma

Author: Jeong Bae Park, Ja Seol Koo

Name of Journal: World Journal of Gastroenterology

**ESPS Manuscript NO:** 5127

The manuscript has been improved according to the suggestions of the reviewers:

1 The format has been updated

- 2 Revision has been made according to the suggestions of the reviewer
- (1) I inserted informative figures regarding endoscopic and histological findings of MALT lymphoma.
- (2) I made changes to Table 1. I modified Table 1 and added a figure (flowchart of diagnosis and treatment)
  - (3) The language in the manuscript has been polished.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Gastroenterology.

Sincerely yours,

188m).

Jeong Bae Park, MD

Division of Gastroenterology

Department of Internal Medicine

Dongguk University Gyeongju Hospital

Dongguk University College of Medicine

87 Dongdae-ro, Gyeongju

780-350, Korea

Tel: +82-54-770-8558

Fax: +82-54-770-8558

E-mail: trueship@medimail.co.kr

